Author's response to reviews

Title: Huge splenic hamartoma in a 39-years old woman and differential diagnosis: a case report

Authors:

Paraskevi Vlachou (vivi_vlachou@hotmail.com)
Dimitris Fagkrezos (fdbl@ath.forthnet.gr)
Anastasia Tzivelopoulou (n.soursos@Soners.gr)
Petros Maniatis (pekatman@otenet.gr)
John Papailiou (johnantpap@ath.forthnet.gr)
Charikleia Triantopoulou (ctriantopoulou@gmail.com)

Version: 2
Date: 6 July 2014

Author's response to reviews: see over
We would like to inform you that we reformatted the title page and we restructured the Abstract section. We think that the case we present is a case which is very rare and will help radiologists interpret their radiologic findings with accuracy. It is also a tool for the clinicians who are interested in these tumors, as oncologists and surgeons, in order to determine the suitable therapeutic approach.